\documentclass[a4paper,10pt]{article}
\usepackage{fullpage}
\usepackage{times}

\begin{document}

\title{L41: Experiment Title Here}
\author{Your Full Name Here \textless{}crsid\textgreater{}}
\date{\today}

\maketitle
\noindent
\textbf{Reminder: The page limit for your lab report is 6 pages.
  (Remove this note before submitting your report.)}

\section{Introduction}

\section{Experimental setup and methodology}

\section{Results and discussion}

\section{Conclusion}

\section{References}

%\section{Optional appendices}

% You might choose to place key DTrace scripts or JupyterLab output here.

\end{document}
